Author's response to reviews

Title: Automated comparison of last hospital main diagnosis and underlying cause of death ICD10 codes, France, 2008-2009

Authors:

Agathe Lamarche-Vadel (agathe.lamarche-vadel@inserm.fr)
Gerard Pavillon (gerard.pavillon@inserm.fr)
Albertine Aouba (albertine.aouba@inserm.fr)
Lars A Johansson (LarsAge.Johansson@socialstyrelsen.se)
Laurence Meyer (laurence.meyer@inserm.fr)
Eric Jougla (Eric.Jougla@inserm.fr)
Gregoire Rey (gregoire.rey@inserm.fr)

Version: 2 Date: 23 July 2013

Author's response to reviews:

Dear Editor,

Please find attached a revised version of our paper entitled "Automated comparison of last hospital main diagnosis and underlying cause of death ICD10 codes, France, 2008-2009".

Following your request, we added a paragraph in the method section of the manuscript about the authorization details for the health register and cause of death data access, including the names of the two data protection committee and institutional ethical review boards which gave approval, as well as the authorization reference numbers.

Best regards,

Grégoire Rey and Agathe Lamarche-Vadel
Inserm - CépiDc
Bâtiment La Force - porte 58
80 rue du Général Leclerc
94270 Kremlin-Bicêtre Cedex
Tel. 33 (1) 49 59 53 38
Fax. 33 (1) 49 59 19 30
Email: agathe.lamarche-vadel@inserm.fr
Site Web : http://www.cepidc.inserm.fr